Author's response to reviews

Title: Acute Angle-Closure Glaucoma in Retinopathy of Prematurity Following Pupil Dilation

Authors:

Shiu-Chen Wu (shiuchen@adm.cgcmh.org.tw)
Yung-Sung Lee (nolosolee@gmail.com)
Wei-Chi Wu (weichi666@gmail.com)
Shirley HL Chang (hlc0625@hotmail.com)

Version: 2 Date: 30 March 2015

Author's response to reviews:

Dear Editors and Reviewers:

We have uploaded a revised manuscript according to BMC Ophthalmology submission guideline and CARE checklist.

The followings are the changes:
1. Add “Case Report” in Title page
2. Add “Consent”, “Competing Interest”, “Author's Contribution”, “Acknowledgments” sections after the Conclusions

Sincerely,
Shirley H.L. Chang, MD
Department of Ophthalmology, Chang Gung Memorial Hospital, Linkou, Taiwan;
5 Fu-Hsin Rd., Kweishan 333, Taoyuan, Taiwan;
E-mail: hlc0625@hotmail.com
TEL: 886-3-3281200 ext.8666;
FAX: 886-3-3287798